Author's response to reviews

Title: Adolescents' perceptions about smokers in Karnataka, India

Authors:

Upendra M Bhojani (upendra@iphindia.org)
Maya A Ellias (maya@iphindia.org)
Narayanan Devadasan (deva@iphindia.org)

Version: 5 Date: 13 July 2011

Author's response to reviews: see over
Dear Editor,

BMC Public Health

Thank you for your suggestions on formatting of our manuscript (MS: 1529668029538609) titled, "A cross-sectional study on adolescents' perceptions about smoker in Karnataka, India.”

We have done all the suggested major and minor format changes in the manuscript along with proof reading. We have uploaded a revised manuscript.

Best regards,

Upendra Bhojani
Maya Annie Elias
N. Devadasan